Author's response to reviews

Title: Sexual behaviour, sexually transmitted infections and attitudes to chlamydia testing among a unique national sample of young Australians: baseline data from a randomised controlled trial

Authors:

Melissa Kang (melissa.kang@sydney.edu.au)
Arlie Rochford (arlie_rochford@hotmail.com)
S Rachel Skinner (rachel.skinner@health.nsw.gov.au)
Adrian Mindel (adrian.mindel@gmail.com)
Marianne Webb (marianne.webb23@gmail.com)
Jennifer Peat (jennypeat@optusnet.com.au)
Tim Usherwood (tim.usherwood@sydney.edu.au)

Version: 2 Date: 22 December 2013

Author's response to reviews:

Please see attached document entitled MS1953710207108680_CoverLetterforRevision_Dec2013.docx